Author's response to reviews

Title: Low back pain status in elite and semi-elite Australian football codes: a cross-sectional survey of football (soccer), Australian-Rules, rugby league, rugby union and non-athletic controls

Authors:

Wayne Hoskins (waynehoskins@iinet.com.au)
Henry Pollard (hpollard@optushome.com.au)
Chris Daff (chirodaff_013@hotmail.com)
Andrew Odell (bubblytang@hotmail.com)
Peter Garbutt (enhancehealthcare@iinet.net.au)
Andrew McHardy (golfinjury@optusnet.com.au)
Kate Hardy (katie.hardy@optusnet.com.au)
George Dragasevic (dragasev@tpg.com.au)

Version: 4 Date: 4 April 2009

Author's response to reviews:

* Removed shading and colours from the tables
* change the layout of the tables so there is no vertical elements
* removed the "et al" if there were ten authors or less in the references and added full names